Global Monsoon Changes with Specific Warming Levels in Two Large-Ensemble Simulations

Zhibo Li¹, Yongyun Hu¹, Ying Sun², Wen Chen³, Ming Sun⁴, Delong Wang⁵, and Yihui Ding⁶

¹Peking University
²National Climate Center, China Meteorological Administration
³Institute of Atmospheric Physics, Chinese Academy of Sciences
⁴Key Laboratory of Meteorological Disaster, Ministry of Education/Joint International Research Laboratory of Climate and Environmental Change/Collaborative Innovation Center on Forecast and Evaluation of Meteorological Disasters, Nanjing University of Information Science and Technology
⁵Nanjing University of Information Science & Technology
⁶National Climate CenterChina Meteorological Administration

November 22, 2022

Abstract

Projection of the global monsoon (GM) system is essential for water resource management, food security, and policymaking. Here we investigate projected changes in global monsoon area (GMA) and global monsoon intensity (GMI) with specific global warming levels, using two datasets of large-ensemble simulations. Both datasets project quasi-linear increases in GMA and GMI with global warming. The GMI over Northern-Hemisphere continents is consistently enhanced, while the GMI over Southern-Hemisphere continents are dominated by opposite changes in the GMI over South America. In addition, both datasets show enhanced monsoon intensity over most parts of regional monsoon domains, except for the North American monsoon. The different changes of the North American monsoon are up on projected temperature differences between the equatorial eastern Pacific and the tropical Atlantic. Moisture budget shows that the thermodynamic component always makes a large positive contribution to the increase in monsoon precipitation, while evaporation has a smaller positive contribution, except for the East Asian monsoon. The contribution of the dynamic component shows large differences for different regional monsoons. Therefore, the different changes in regional monsoon precipitation are mainly caused by the dynamic component.

Hosted file

essoar.10510856.1.docx available at https://authorea.com/users/537671/articles/599401-globalmonsoon-changes-with-specific-warming-levels-in-two-large-ensemble-simulations

Global Monsoon Changes with Specific Warming Levels in Two Large-Ensemble Simulations

Zhibo Li¹, Yongyun Hu^{1*}, Ying Sun^{2,3}, Wen Chen^{4,5}, Ming Sun³, Delong Wang⁶, Yihui Ding²

¹ Laboratory for Climate and Atmosphere - Ocean Studies, Department of Atmospheric and Oceanic Sciences, School of Physics, Peking University, Beijing 100871, China

² National Climate Center, Laboratory for Climate Studies, China Meteorological Administration, Beijing 100081, China

³ Collaborative Innovation Center on Forecast and Evaluation of Meteorological Disasters (CIC-FEMD), Nanjing University of Information Science and Technology, Nanjing 210044, China

⁴ Center for Monsoon System Research, Institute of Atmospheric Physics, Chinese Academy of Sciences, Beijing 100029, China

⁵ College of Earth and Planetary Sciences, University of Chinese Academy of Sciences, Beijing 100049, China

⁶ Beijing Institute of Applied Meteorology, Beijing 100029, China

To be submitted to JGR-Atmosphere

*Corresponding author: Yongyun Hu (yyhu@pku.edu.cn)

Key points:

- Global monsoon area and intensity both show quasi-linear increasing with global warming levels.
- Projection of North American monsoon depend on temperature differences between the equatorial eastern Pacific and the tropical Atlantic.
- Dynamic component plays the dominant role in causing changes in monsoon precipitation, thermodynamic component makes positive contributions.

Abstract

Projection of the global monsoon (GM) system is essential for water resource management, food security, and policymaking. Here we investigate projected changes in global monsoon area (GMA) and global monsoon intensity (GMI) with specific global warming levels, using two datasets of large-ensemble simulations. Both datasets project quasi-linear increases in GMA and GMI with global warming. The GMI over Northern-Hemisphere continents is consistently enhanced, while the GMI over Southern-Hemisphere continents are dominated by opposite changes in the GMI over South America. In addition, both datasets show enhanced monsoon intensity over most parts of regional monsoon domains, except for the North American monsoon. The different changes of the North American monsoon are up on projected temperature differences between the equatorial eastern Pacific and the tropical Atlantic. Moisture budget shows that the thermodynamic component always makes a large positive contribution to the increase in monsoon precipitation, while evaporation has a smaller positive contribution, except for the East Asian monsoon. The contribution of the dynamic component shows large differences for different regional monsoons. Therefore, the different changes in regional monsoon precipitation are mainly caused by the dynamic component.

Plain Language Summary

Future changes in the global monsoon (GM) have a great impact on human society, water resource management, food security, and policymaking. Using two large-ensemble simulations, we study the changes in global monsoon with specific global mean surface temperature increases, such as $1.5 \,^{\circ}$ C to $5 \,^{\circ}$ C of warming. We find that the global monsoon area and monsoon precipitation intensity will quasi-linearly increase in the future. The global monsoon intensity over Northern-Hemisphere land will be consistently enhanced, while the opposite changes over Southern-Hemisphere land. The intensity changes in the North American monsoon will be up on temperature difference between the equatorial eastern Pacific and the tropical Atlantic in the future. The thermodynamic component always makes a large positive contribution to increasing monsoon rainfall, evaporation only has significant positive effect on the East Asian monsoon, and the dynamic component is most important in determining monsoon precipitation changes in the future.

1 Introduction

Over two-thirds of the global population is influenced by the global monsoon (GM) system, and the GM variability is of essential scientific and social-economic importance (Wang and Ding 2008). During the past two decades, how GM changes in future, in responding to global greenhouse warming, has been a crucial issue for food security, water resource management, and policymaking.

There have been extensive works that investigated GM projections, using Coupled Model Intercomparison Project (CMIP) datasets. Based on CMIP phase 3 (CMIP3), Kim et al. (2008) found that multi-model ensemble means can capture GM precipitation, but model spreads are large. Hsu et al. (2012, 2013) found increases in GM area (GMA) and GM precipitation (GMP) in both the CMIP3 and CMIP5 projection simulations. They attributed it to increases in moisture convergence and surface evaporation. Lee and Wang (2014) showed that CMIP5 models have improved abilities in simulating GMP. They pointed out that Northern-Hemisphere (NH) GMP will increase due to an increase in temperature contrast between the NH and Southern Hemisphere (SH), and atmospheric moistening, against the enhanced troposphere atmospheric stability. The Sixth Assessment Report of the Intergovernmental Panel on Climate Change (IPCC-AR6) showed that the GMP will increase in the mid-to long term future, particularly over South and Southeast Asia, East Asia, and West

Africa (IPCC, 2021).

Recently, CMIP6 simulations are used to study future GM changes under global warming. Lee et al. (2019) showed a more significant increase in monsoon precipitation over land than over ocean. Moon and Ha (2020) found lengthened rainy season over most parts of monsoon domains, except for the North and South American monsoons. Wang et al. (2020a) showed that NH monsoon precipitation will increase, whereas SH monsoon precipitation has almost no changes. The projected stronger inter-hemispheric thermal contrast enhanced the NH monsoon (Wang et al. 2020b).

Most of the above works on GM projections are based on multi-model ensembles of transient climate simulations. Considering the problems arising from different climate sensitivities and inconsistent definitions of the global warming levels, Deser et al. (2020) emphasized the advantage of initial-condition large-ensemble simulations of individual climate models. Large-ensemble simulations ensure enough members to investigate climate changes due to greenhouse gas forcing under specific global warming levels, which is one of the central issues in the current international negotiation of climate changes, including the Paris Agreement (Mitchell et al. 2017; Milinski et al. 2020; Zhou et al. 2020).

In this work, we use two state-of-art large-ensemble simulations to investigate GM changes and the relationship between GM indices and global warming levels. We try to find both the similarity and difference between the two large-ensemble simulations and compare the results with those from CMIP6 models. We only focus on the contribution of CO_2 forcing to GM changes, while contributions from natural variabilities are not considered in this work.

The paper is organized as follows: The data and methodology are described in Section 2. Section 3 evaluates the GM simulation ability of two datasets. Future changes in the GM, regional monsoons (RM), and associated physical mechanisms are presented in Section 4. Conclusions are summarized in Section 5.

2 Data and Methods

2.1 Observational and Model Data

The observational datasets used in this study include monthly gridded precipitation from the Global Precipitation Climatology Project (GPCP, 1979-2016; Adler et al. 2003) and the Center for Climate Prediction Merged Analysis of Precipitation (CMAP, 1979-2016; Xie & Arkin, 1997) with a resolution of $2.5^{\circ} \times 2.5^{\circ}$ in latitude and longitude.

We use two datasets of large-ensemble simulations. The datasets have been well used in studying monsoon changes under global warming (Li et al. 2019, 2021a, 2021b; Huang et al. 2020; Zhou et al. 2020). The first large-ensemble simulations are by the Canadian Earth System Model version 2 (CanESM2; Arora et al. 2011). CanESM2 large-ensemble simulations expanded each of the 5 ensemble members from the CMIP5 into 10 members, and to form 50-member

large-ensemble simulations started with slightly different initial conditions (Fyfe et al. 2017). The 50 simulations are driven by historical anthropogenic and natural external forcing for 1950-2005, the Representative Concentration Pathways (RCP) 8.5 scenario for 2006-2100, with horizontal resolution of T42 (approximately 2.8°). We also use the 5 members from the CMIP5 to obtain the global mean surface temperature (GMST) changes relative to the preindustrial period.

The second large-ensemble simulations are from the Max Planck Institute Earth System Model (MPI-ESM; Maher et al. 2019). It is an updated version of MPI-ESM low resolution from the CMIP5, with horizontal resolution of T63 (approximately 1.9°). Similar to CanESM2, individual ensemble members of MPI-ESM only differ in their initial conditions. The historical simulation period of MPI-ESM is 1850-2005 and the RCP8.5 simulations are performed for 2006-2099. All the datasets were converted to $2^{\circ} \times 2^{\circ}$ in latitude and longitude to ensure that they are comparable.

2.2 Global Monsoon Indices and Regional Land Monsoon Domains

Following Wang and Ding (2006, 2008) and Hsu et al. (2011), the GMA is defined as the area in which the local summer-minus winter precipitation (annual range) exceeds 2 mm/day, and the local summer precipitation exceeds 55% of annual total rainfall. In the NH, summer is defined as May-September (MJJAS), and winter is defined as November-March (NDJFM). The definition for the SH is the opposite. The GMP is defined as total summer (June-August for the NH and December-February for the SH) precipitation over the GMA. And the GM intensity (GMI) is defined as the GMP amount per unit area, which means the GMP is divided by the GMA. Because the area of each grid box varies with latitude, we used an area-weight method to calculate these indices.

The GM domain is separated into seven regional land monsoon domains as the Figure 1 in Jin et al. (2020). The seven regional land monsoon domains are East Asia (EA; 105 °E-140 °E, 22.5 °N-65 °N), South Asia (SA; 60 °E-105 °E, 0-40 °N), North Africa (NAF; 20 °W-50 °E, 0-20 °N), South Africa (SAF; 0-60 °E, 40 °S-0), North America (NAM; 120 °W-50 °W, 0-30 °N), South America (SAM; 85 °W-30 °W, 40 °S-0), and Australia (AUS; 100 °E-155 °E, 30 °S-0), respectively. Note that the NAM domain covers not only Mexico but also Central America and Venezuela.

2.3 Global Warming Levels in Two Models

Following Li et al. (2019) and Zha et al. (2021), we use two baseline periods to estimate future changes in GM indices and their relationships with different global warming levels. The first baseline period is the preindustrial period of 1850-1900, which is used to calculate GMST changes relative to the preindustrial period. The second baseline period is the current climate, which is used to compare future changes in GM indices. We define the 1 °C warming period in CanESM2 and MPI-ESM as the current climate, since the observed GMST in 2018-2019 was about 1 °C warming relative to the preindustrial period (WMO 2020). Hence, the six 10-year intervals in CanESM2 are: 1995-2004 (0.983 °C),

2007-2016 (1.454 °C), 2021-2030 (1.981 °C), 2042-2051 (2.958 °C), 2061-2070 (3.978 °C), and 2079-2088 (4.980 °C). The five 10-year intervals in MPI-ESM are: 1995-2004 (1.017 °C), 2015-2024 (1.499 °C), 2031-2040 (1.996 °C), 2055-2064 (2.990 °C), and 2077-2086 (4.016 °C). Hereafter, GMST changes are all relative to the preindustrial period, and changes in GM indices are relative to the current climate (1 °C of warming).

2.4 Method of Moisture Budget

Jin et al. (2020) used a method of simplified moisture budget to analyze the relative contributions of dynamic and thermodynamic components to monsoon precipitation changes:

(1)

Where P denotes the model-output precipitation, denotes the vertical velocity at 500 hPa, denotes the specific humidity at 850 hPa, and g denotes the acceleration of gravity (9.8 m/s²). This formula is simplified from the complete moisture budget formula, and it has the ability to directly demonstrate the dynamic (vertical motion) and thermodynamic (water vapor) components of the change in monsoon precipitation (see Figure 14 in Jin et al. 2020).

Similar simplified budget decomposition was also found to be sufficient in analyzing tropical and monsoon precipitation changes (Huang et al. 2013; Sooraj et al. 2015; Li et al. 2019; Li et al. 2021a). Because surface evaporation may affect the GMP change (Jin et al. 2020; Wang et al. 2020b), diagnosis equations for projecting the future GMP can be written as:

- (2)
- (3)
- (4)

The overbar and denote the values for the current climate and the difference between future and current climates, respectively. The first term on the righthand side of Equation 2 indicates the dynamic component (DY), and the second term indicates the thermodynamic component (TH). E denotes evaporation changes. Each item in Equation 2 is calculated at grid points and is areaweighted according to the evolving GMA. Detailed information can be found in Jin et al. (2020).

Besides, locally estimated scatterplot smoothing is used for scatterplot smoothing (LOESS; Cleveland 1979). Two methods are used to examine the significance of GM changes under specific global warming levels. The first is the student t-test at the 0.05 confidence level. The second is defined as over 75% of the results from ensemble members that show consistent changes.

3 Model Evaluation

Firstly, model-simulated 30-year (1977-2006 for CanESM2, 1971-2000 for MPI-ESM) mean precipitation and GM indices are compared with the observations

(1986-2015). GMST changes over the three intervals are all approximately 0.69°C above the preindustrial level. Figure 1 shows the climatological annualmean precipitation of observations and the ensemble-member mean from the two models. Both models well reproduce the observed annual-mean precipitation in tropical and subtropical regions. The spatial correlation coefficient between observations and CanESM2 and MPI-ESM is 0.85 and 0.79, respectively. However, both models have poor performance in simulating precipitation over the South American continent. This could be because the models underestimate the variability of the SAM monsoon and its sensitivity to climate forcing (Fu et al. 2013). Besides, the double ITCZ problem causes excessive precipitation in the south of the equator, which is a common problem in coupled general circulation models (Zhou et al. 2008; Li and Xie 2014).

As shown in Figure 2, both models well reproduce the climatological annual range of precipitation (shading) and GMA (black contour). The spatial correlation coefficient of the annual range of precipitation between observations and CanESM2 (MPI-ESM) is 0.80 (0.79). The poor performance in simulating GMA over the tropical south Pacific coincides with the spatial pattern of the simulated double ITCZ. As shown in Table 1, the observed standard deviations (STDs) are 46.8^{*10^9} m³/day, 5.43^{*10^6} km², and 0.16 mm/day for GMP, GMA, and GMI, respectively. The simulated STDs of the three GM indices are calculated for ensemble-mean from the historical experiments, and the median values and their 5-95% ranges are estimated. The simulated STDs of three GM indices are 41.3 (32.7-51.7) $*10^9$ m³/day, 4.48 (3.53-5.83) $*10^6$ km², and 0.15 (0.12-0.19) mm/day for CanESM2, respectively. They are 33.8 $(27.6-45.2) *10^9$ m^3/day , 3.76 (2.98-4.84) *10⁶ km², and 0.16 (0.13-0.19) mm/day for MPI-ESM, respectively. The observed STDs are generally located in the 5-95% ranges, which indicates the good performance of the two models in reproducing interannual GM variabilities. The above evaluations give us confidence for using these two large-ensemble simulations to estimate GM changes under specific global warming levels.

4 Projection Changes in Global Monsoon

4.1 Relationship between GM and global warming Levels

Figure 3 shows changes in the annual range of precipitation of CanEMS2 when the GMST is increased from 1.5 °C to 5 °C above the preindustrial levels. The change of the annual range is not significant for 1.5 °C of warming in most parts of monsoon domains (Figure 3a). When the warming level reaches 2 °C, however, negative annual ranges over the American continent and positive annual ranges over the NAF become statistically significant (Figure 3b). As GMST is increased more, anomalous values of the annual range become stronger and more significant (Figures 3c-e). The annual range increases over most monsoon domains, except for the NAM and SAM. It indicates a weakening of the NAM monsoon and SAM monsoon in the future. There is a large positive annual range over the near-equator Pacific, which indicates an equatorward movement of the ITCZ and the monsoon convergence zone. Furthermore, the ITCZ movement causes reduced precipitation over large areas of the NAM (Wang et al. 2020a). The changes in GMA well cooperate with the annual range changes, GMA tends to expand (shrink) at the regions where seasonal differences of precipitation are enhanced (weakened), especially at the flank of the monsoon domain. The GMA expands toward high latitudes due to amplified precipitation difference between summer and winter in a warmer world.

Similar to the results in CanESM2, the results in MPI-ESM also show enhanced annual range over Asia, Africa, and Australia (Figures 4a-d). However, the annual range of the SAM monsoon shows a significant increase, which is in contrast with that in CanESM2 (Figure 3). As mentioned in Section 3, both models cannot reproduce the climatological annual range of the SAM monsoon. The inter-model discrepancies for the SAM have been noticed in the CMIP5 models (Fu et al. 2013; Yin et al. 2013), and the risk of strong climatic drying and potential rainforest die-back in the future remains a great concern (Wang et al. 2020b). The above model-dependent results over the NAM and SAM remind us that the analysis of model uncertainty and inter-model difference is necessary for future projections.

Figure 5 shows scatter plots of GMA and GMI indices against GMST changes in the two models. All LOESS regressions of the two GM indices show an increase with global warming levels and the values of the GM indices in MPI-ESM are larger than that in CanESM2. The model ensemble-mean of GMA and GMI will approximately increase from 95 to 107 $*10^6$ km² and from 7.10 to 7.65 mm/day through the 150 years for CanESM2, respectively, while they are from 110 to 114 $*10^6$ km² and from 7.55 to 7.90 mm/day for MPI-ESM, respectively. The increasing rates of GM indices in CanESM2 are larger than that in MPI-ESM. The simulation spreads for the two models are similar, which are around 16 $*10^6$ km² and 1 mm/day for GMA and GMI, respectively. The comparable values between long-term changes and simulation spreads in the GM indices also indicate the contributions from natural variability to the GM changes are large in the near future (Deser et al. 2020; Huang et al. 2020; Zhou et al. 2020; Li et al. 2021b).

We further analyze the probability density functions (PDFs) of GMA and GMI in the two models (Figure 6). The PDFs of the GM indices shift rightward with global warming levels, indicating increases in the mean values of GMA and GMI. In addition, the GMI (Figures 6c-d) in both models shift rightward and spread wider, indicating increases in both the mean GMI and its variabilities with global warming. These results further confirm enhanced frequency and intensity of extreme events of monsoon rainfall in the future (Wang et al. 2020b).

4.2 Changes of GM indices per degree of warming

In this subsection, we quantify the relationship between GM indices and global warming. The change in a GM index over the 95th (75th) percentage is defined as a strong (weak) change. The GMI in CanESM2 shows quasi-linear increasing (~1 %/°C) with global warming. The GMA shows a larger expansion than the

GMI at lower warming levels (Figure 7a left). Though the increase in GMA of MPI-ESM shows smaller increases compared with that of CanESM2, the GMI shows similar quasi-linear increases around 1 %/°C (Figure 7a right).

Furthermore, we calculate the changes in GM indices over global land, NH land, and SH land (Figures 7b-d), respectively. The GMI over global land generally shows weakly quasi-linear increases (~1 %/°C) in CanESM2, while strongly quasi-linear increases (~2 %/°C) in MPI-ESM. The uncertainty of changes in global land monsoon precipitation may arise from atmospheric circulation changes, which is partly due to model-dependent responses to uniform SST warming (Chen et al. 2020).

Both models show strong enhancement in the GMI over the NH land. The GMI increases from $1 \%/^{\circ}$ C to $3 \%/^{\circ}$ C in CanESM2. For MPI-ESM, the GMI shows a quasi-linear increase (~2.2 %/^{\circ}C) in the future (Figure 7c). This robust change in the GMI over the NH land indicates more extreme events of monsoon precipitation (Wang et al. 2020b). As shown in Figure 7d, there are large diversities of GM indices over the SH land. For CanESM2, there are barely no changes in GMI, and the simulation spreads are large. For MPI-ESM, the GMI show quasi-linear and strong increases with global warming. The unmatched changes in the GM indices over the SH land suggest that the associated physical mechanisms need to be studied in future work, such as contributions from model frameworks and natural variabilities (Deser et al. 2020; Huang et al. 2020). Besides, it is interesting to note that the increasing rate of the GMA over the NH land shows a non-linear weakening with global warming in both models, which indicates that the expanding trend in the GMA will slow down in a warmer world.

Based on the two large-ensemble simulations, changes in regional monsoon indices are shown in Figure 8. For the EA (Figure 8a), both models show large increases in the regional monsoon intensity (RMI). The two models show differences in the regional monsoon area (RMA), with strong expansion in CanESM2 and weak expansion in MPI-ESM. Li et al. (2019) proposed that rapid increasing in monsoon precipitation may be related to the positive feedback between the monsoon circulation and precipitation. For the SA (Figure 8b), the RMI shows strong increasing when the GMST warming is greater than 2 °C, indicating the important impact of the 2 °C thresholds on people living in the SA.

For the NAF (Figure 8c), the increasing trend in RMI is non-linearly weakened with warming levels. The RMA expands mainly northward (Figures 3 and 4), leading to the wetter Sahel. For the SAF (Figure 8d), the RMI shows consistent increases with warming levels, while RMA changes have large uncertainties.

There are large diversities of changes in the NAM and SAM (Figures 8e and 8f), as noticed in Section 3. Both RMA and RMI of the NAM show decreases in CanESM2, while they have little change in MPI-ESM (Figures 8e). Figure 8f shows that the SAM RMI quasi-linearly decreases (increases) in CanESM2 (MPI-ESM). However, the RMA has no significant changes. For the AUS (Figure

8g), the RMI does not change very much because of the large spreads of the RMA in large-ensemble simulations.

In summary, the results in RMI changes over EA, SA, NAF, SAF, and AUS are similar in the two models, and the changes are similar to those based on CMIP6 projections (Chen et al. 2020; Jin et al. 2020; Wang et al. 2020b). In contrast, the changes in RMA and RMI of NAM and SAM are model-dependent.

CMIP6 ensemble projects a drier NAM and a non-significant wetter SAM (Wang et al. 2020b). Following Wang et al. (2020a), we find the difference of surface air temperatures (SAT) between the equatorial eastern Pacific (5 °S-5 °N, 120 °W-80 °W) and the tropical Atlantic (10°N-20 °N, 60 °W-15 °W) increase in both models with warming levels (Figures 9a and 9b). The SAT difference between the two oceans is associated with the movement of ITCZ. When the warming rate of the equatorial eastern Pacific is faster than that of the tropical Atlantic, the El Niño-like warming results in an equatorward shift of the ITCZ and the monsoon convergence zone, which consequently leads to reduced precipitation of the NAM. This relationship is reproduced in both models (Figures 9c and 9d), with a much greater trend in CanESM2 $(-1.1 \text{ mm/day})^{\circ}$ C) than in MPI-ESM (-0.17 mm/day/°C). The SAT difference in CanESM2 increases from -0.25 °C to 0.85 °C, while it changes little in MPI-ESM. It suggests that global warming has a stronger effect on the NAM changes in CanESM2. In addition, the differences of RMA and RMI may partly be due to the too small domain of the NAM monsoon. As for the RMI changes over the SAM, the results from CMIP5 (Yin et al. 2013) and CMIP6 models (Jin et al. 2020) are model-dependent.

4.3 Moisture Budget of the Global Monsoon Precipitation

Changes in atmospheric water vapor, circulation, and evaporation all would contribute to future changes in monsoon precipitation. Using the method of the simplified moisture budget (Equation 2), we estimate monsoon precipitation changes over global land and seven RM domains, respectively. It is found that the TH component always has positive contributions to precipitation increases under global warming, and the inter-member standard deviation is small. It indicates that global warming would lead to a wetter atmosphere and consequently more monsoon precipitation.

It is found that E plays an important positive role in EA monsoon precipitation, a weaker positive role in SA and AUS monsoon precipitation, and a negligible role for other RM domains. The enhanced surface evaporation is mainly caused by the surface temperature increasing, which is a part of the thermodynamic effects (Fasullo 2012). The standard deviations of E are similar to that of TH. In contrast, DY shows very different changes over different regional monsoon domains and between the two models, and its standard deviations are much larger. Our results are consistent with the argument by Wang et al. (2020a), that is, DY plays a more important role than TH in causing changes in monsoon precipitation. DY shows similar changes over SA (Figure 10c) and SAF (Figure 10e) monsoon domains in both models. However, DY makes a positive contribution to monsoon precipitation increases over EA (Figure 10b), NAF (Figure 10d), and AUS (Figure 10h) in CanESM2, whereas DY has little change over these RM domains in MPI-ESM. DY has dominant and negative contributions to monsoon precipitation increases over NAM (Figure 10f) and SAM (Figure 10g) in CanESM2, while it has almost no changes in MPI-ESM.

The uncertainty of monsoon circulations associated with DY changes is closely related to the projected warming pattern and also related to model biases, such as the cold-tongue bias in tropical eastern Pacific (Li and Xie 2014), cloud and water vapor feedbacks (Jalihal et al. 2019), land-atmosphere interaction (Wang et al. 2020b), and so on.

5 Conclusions and Discussion

We use large-ensemble simulations with CanESM2 and MPI-ESM to investigate GM changes associated with global greenhouse warming. Both large-ensemble simulations well reproduce the climatological large-scale features of the GM, except for the poor performance in simulating SAM monsoon. When global warming is over 2 °C, the annual range of monsoon precipitation consistently increases in most parts of the Asian-Australian and African monsoon domains in both models, while the annual range over the NAM and SAM shows opposite changes in the two models. CanESM2 projects a much weaker annual range of the NAM and SAM, while MPI-ESM yields a stronger annual range of precipitation for the SAM and little change for the NAM.

GMA and GMI increase with global warming levels. The increasing rates in CanESM2 are larger than that in MPI-ESM. The GMI increases (~1 %/°C) in both the climatological mean and internal variability. It indicates increasing of frequency and intensity of extreme events of monsoon rainfall. Both models demonstrate consistent enhancement in the GMI over the NH land, which is likely due to the enhanced NH-SH thermal contrast (Lee and Wang 2014; Wang et al. 2020a). For the GM indices over the SH land, the two models generate different projections, and the ensemble-member spreads are large. This is largely due to the combined effects of model biases and internal variabilities of the NAM and SAM.

For RM changes, both models generate similar enhanced RMI of the EA (strong), SA (strong), NAF (weak), SAF (weak), and AUS (slight), which is also similar to the CMIP6 projections (Chen et al. 2020; Jin et al. 2020; Wang et al. 2020b). In contrast, the two models yield different projections of RMA and RMI for the NAM and SAM. The NAM and SAM become drier and weaker in CanESM2, while SAM becomes wetter and stronger in MPI-ESM. RMA and RMI of the NAM have almost no change in MPI-ESM. The discrepancy in NAM monsoon projection between the two models is because of different model projections of SAT differences between the equatorial eastern Pacific and the tropical Atlantic. CanESM2 simulates stronger warming over the equatorial eastern Pacific, which is comparable to the results from CMIP6 ensemble-mean (Wang et al. 2020a). And the El Niño-like warming causes an equatorward shift of the ITCZ, which leads to weakened NAM monsoon. In contrast, SAT difference between the two oceans is weaker in MPI-ESM projections. The model-dependence of NAM and SAM projections needs to be studied with comparison from more models.

The relative contribution of dynamic and thermodynamic components to changes in global and regional land monsoon precipitation are estimated, using a simplified method of moisture budget. The thermodynamic component has a positive contribution to monsoon precipitation increases for all regional monsoons in both models, because of the increase in specific humidity due to greenhouse warming. In contrast, the dynamic component has very different contributions to changes in regional monsoon precipitation, and both model discrepancies and inter-member standard deviation are large. Therefore, the different changes in regional monsoon precipitation are caused by the dynamic component. This is because global greenhouse warming leads to different changes in regional monsoon circulations.

Conflict of Interests

The authors declare no conflicts of interest.

Acknowledgments

This work is supported by the National Natural Science Foundation of China under grant 41888101. CanESM2 data are available from the government of Canada's open data portal https://open.canada.ca/data/en/dataset/aa7b6823-fd1e-49ff-a6fb-68076a4a477c. MPI-ESM grand ensemble data are available for noncommercial use via https://esgf-data.dkrz.de/projects/MPI-ESM/.

References

Adler, R. F., Huffman, G. J., Chang, A., Ferraro, R., Xie, P. P., Janowiak, J., et al. (2003). The version 2 Global Precipitation Climatology Project (GPCP) monthly precipitation analysis (1979-present). *Journal of Hydrometeorology*, 4(6), 1147-1167. https://doi.org/10.1175/1525-7541(2003)004

Arora, V. K., Scinocca, J. F., Boer, G. J., Christian, J. R., Denman, K. L., Flato, G. M., et al. (2011). Carbon emission limits required to satisfy future representative concentration pathways of greenhouse gases. *Geophysical Research Letters*, 38, L05805. https://doi.org/10.1029/2010GL046270

Chen, Z., Zhou, T., Zhang, L., Chen, X., Zhang, W., & Jiang, J. (2020). Global land monsoon precipitation changes in CMIP6 projections. *Geophysical Research Letters*, 47, e2019GL086902. https://doi.org/10.1029/2019GL086902

Cleveland, W. S. (1979). Robust locally weighted regression and smoothing scatterplots. *Journal of the American Statistical Association*, 74(368), 829-836. https://doi.org/10.2307/2286407

Deser, C., Lehner, F., Rodgers, K. B., Ault, T., Delworth, T. L., DiNezio, P. N., et al. (2020). Insights from Earth system model initial-condition large

ensembles and future prospects. Nature Climate Change, 10(4), 277-286. https://doi.org/10.1038/s41558-020-0731-2

Fasullo, J. (2012). A mechanism for land-ocean contrasts in global monsoon trends in a warming climate. Climate Dynamics, 39, 1137-1147. https://doi.org/10.1007/s00382-011-1270-3

Fu, R., Yin, L., Li, W., Arias, P. A., Dickinson, R. E., Huang, L., et al. (2013). Increased dry-season length over southern Amazonia in recent decades and its implication for future climate projection. *Proceedings of the National Academy* of Sciences, 110(45), 18110-18115. https://doi.org/10.1073/pnas.1302584110

Fyfe, J. C., Derksen, C., Mudryk, L., Flato, G. M., Santer, B. D., Swart, N. C., et al. (2017). Large near-term projected snowpack loss over the western United States. *Nature Communications*, 8(1), 14, 996. https://doi.org/10.1038/ncomms14996

Hsu, P.-C., Li, T., & Wang, B. (2011). Trends in global monsoon area and precipitation over the past 30 years. *Geophysical Research Letters*, 38, L08701. https://doi.org/10.1029/2011GL046893

Hsu, P.-C., Li, T., Luo, J. J., Murakami, H., Kitoh, A., & Zhao, M. (2012). Increase of global monsoon area and precipitation under global warming: A robust signal? *Geophysical Research Letters, 39*, L06701. https://doi.org/10.1029/2012GL051037

Hsu, P.-C., Li, T., Murakami, H., & Kitoh, A. (2013). Future change of the global monsoon revealed from 19 CMIP5 models. *Journal of Geophysical Research: Atmospheres*, 118(3), 1247-1260. https://doi.org/10.1002/jgrd.50145

Huang, P., Xie, S. P., Hu, K., Huang, G., & Huang, R. (2013). Patterns of the seasonal response of tropical rainfall to global warming. *Nature Geoscience*, 6(5), 357–361, https://doi.org/10.1038/NGEO1792

Huang, X., Zhou, T., Dai, A., Li, H., Li, C., Chen, X., et al. (2020). South Asian summer monsoon projections constrained by the Interdecadal Pacific Oscillation. *Science Advances*, 6(11), eaay6546. https://doi.org/10.1126/sciadv.aay6546

IPCC, 2021: Summary for Policymakers. In: Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change [Masson-Delmotte, V., P. Zhai, A. Pirani, S. L. Connors, C. Péan, S. Berger, N. Caud, Y. Chen, L. Goldfarb, M. I. Gomis, M. Huang, K. Leitzell, E. Lonnoy, J.B.R. Matthews, T. K. Maycock, T. Waterfield, O. Yelekci, R. Yu and B. Zhou (eds.)]. Cambridge University Press. In Press.

Jalihal, C., Srinivasan, J., & Chakraborty, A. (2019). Modulation of Indian monsoon by water vapor and cloud feedback over the past 22,000 years. *Nature Communications*, 10, 5071. https://doi.org/10.1038/s41467-019-13754-6

Jin, C., Wang, B., & Liu, J. (2020). Future changes and controlling factors of

the eight regional monsoons projected by CMIP6 models. *Journal of Climate*, 33(21), 9307-9326. https://doi.org/10.1175/JCLI-D-20-0236.1

Kim, H.-J., Wang, B., & Ding, Q. (2008). The global monsoon variability simulated by CMIP3 coupled climate models. *Journal of Climate*, 21(20), 5271-5294. https://doi.org/10.1175/2008JCLI2041.1

Lee, J.-Y., & Wang, B. (2014). Future change of global monsoon in the CMIP5. Climate Dynamics, 42(1-2), 101-119. https://doi.org/10.1007/s00382-012-1564-0

Lee, J.-Y., Yun, K.-S., Yang, Y.-M., Chung, E.-S., & Babu, A. (2019). Challenges in contrasting future change of global land precipitation in CMIP6 models. Presented at the WMO Workshop on Monsoon Climate Change Assessment, 2-3 December, Zhuhai, China.

Li, G., & Xie, S.-P. (2014). Tropical biases in CMIP5 multimodel ensemble: The excessive equatorial Pacific cold tongue and double ITCZ problems. *Journal of Climate*, 27(4), 1765-1780. https://doi.org/10.1175/JCLI-D-13-00337.1

Li, T., Wang, Y., Wang, B., Ting, M., Ding, Y., Sun, Y., et al. (2021). Distinctive South and East Asian Monsoon circulation responses to global warming. *Science Bulletin*, online. https://doi.org/10.1016/j.scib.2021.12.001

Li, Z., Sun, Y., Li, T., Chen, W., & Ding, Y. (2021a). Projections of South Asian summer monsoon under global warming from 1.5° to 5°C. *Journal of Climate*, 34(19), 7913-7926. https://doi.org/10.1175/JCLI-D-20-0547.1

Li, Z., Chen, W., Chen, S., Sun, Y., & Qian, D. (2021b). Uncertainty of central China summer precipitation and related natural internal variability under global warming of 1°C to 3°C. *International Journal of Climatology*, 41(15), 6640-6653. https://doi.org/10.1002/joc-20-0740

Li, Z., Sun, Y., Li, T., Ding, Y., & Hu, T. (2019). Future changes in East Asian summer monsoon circulation and precipitation under 1.5 to 5 °C of warming. *Earth's Future*, 7(12), 1391-1406. https://doi.org/10.1029/2019EF001276

Maher, N., Milinski, S., Suarez-Gutierrez, L., Botzet, M., Dobrynin, M., Kornblueh, L., et al. (2019). The Max Planck Institute Grand Ensemble: Enabling the Exploration of Climate System Variability. *Journal of Advances in Modeling Earth Systems*, 11, 1-21. https://doi.org/10.1029/2019MS001639

Milinski, S., Maher, N., & Olonscheck. (2020). How large does a large ensemble need to be? *Earth System Dynamics*, 11(4), 885-901. https://doi.org/10.5194/esd-11-885-2020

Mitchell, T. D., & Jones, P. D. (2017). An improved method of constructing a database of monthly climate observations and associated highresolution grids. *International Journal of Climatology*, 25(6), 693-712. https://doi.org/10.1002/joc.1181 Moon, S., & Ha, K.-J. (2020). Future changes in monsoon duration and precipitation using CMIP6. *npj Climate and Atmospheric Science*, 3(1), 45. https://doi.org/10.1038/s41612-020-00151-w

Sooraj, K. P., Terray, P., & Mujumdar, M. (2015). Global warming and the weakening of the Asian summer monsoon circulation: assessments from the CMIP5 models. *Climate Dynamics*, 45(1-2), 233-252, https://doi.org/10.1007/s00382-014-2257-7

Wang, B., & Ding, Q. (2006). Changes in global monsoon precipitation over the past 56 years. *Geophysical Research Letters*, 33, L06711. https://doi.org/10.1029/2005GL025347

Wang, B., & Ding, Q. (2008). Global monsoon: dominant mode of annual variation in the tropics. *Dynamics of Atmospheres and Oceans*, 44(3-4), 165-183. https://doi.org/10.1016/j.dynatmoce.2007.05.002

Wang, B., Jin, C., & Liu, J. (2020a). Understanding future change of global monsoon projected by CMIP6 models. *Journal of Climate*, 33(15), 6471-6489. https://doi.org/10.1175/JCLI-D-19-0993.1

Wang, B., Biasutii, M., Byrne, M. P., Castro, C., Chang, C.-P., Cook, K., et al. (2020b). Monsoons climate change assessment. *Bulletin of the American Meteorological Society*. https://doi.org/10.1175/BAMS-D-19-0335.1

Xie, P., & Arkin, P. A. (1997). Global precipitation: A 17-year monthly analysis based on gauge observations, satellite estimates and numerical model outputs. *Bulletin of the American Meteorological Society*, 78(11), 2539-2558. https://doi.org/10.1175/1520-0477(1997)078

Yin, L., Fu, R., Shevliakova, E., & Dickinson, R. E. (2013). How well can CMIP5 simulate precipitation and its controlling processes over tropical South America? *Climate Dynamics*, 41(11-12), 3127-3143. https://doi.org/10.1007/s00382-012-1582-y

Zha, J., Shen, C., Li, Z., Wu, J., Zhao, D., Fan, W., et al. (2021). Projected changes in global terrestrial near-surface wind speed in 1.5°C-4.0°C global warming levels. *Environmental Research Letters*, 16, L114016. https://doi.org/10.1088/1748-9326-ac2fdd

Zhou, T., Yu, R., Li, H., & Wang, B. (2008). Ocean forcing to changes in global monsoon precipitation over the recent half-century. *Journal of Climate*, 21(15), 3833-3852. https://doi.org/10.1175/2008jcli2067.1

Zhou, T., Lu, J., Zhang, W., & Chen, Z. (2020). The sources of uncertainty in the projection of global land monsoon precipitation. *Geophysical Research Letters*, 47, e2020GL088415. https://doi.org/10.1029/2020GL088415

Table captions

Table 1. The standard deviation (STDs) of global monsoon precipitation (GMP, m^3/day), global monsoon area (GMA, km^2), and global monsoon intensity (GMI, mm/day) from CanESM2 (1977-2006), MPI-ESM (1971-2000), and average of GPCP and CMAP (1986-2015). The STDs of the models are the median values from all members, and the brackets are the 5-95% range of the member spread.

Figure captions

Figure 1. Climatological and annual mean precipitation (mm/day). (a) GPCP and CMAP precipitation averaged 1986-2015, (b) 50 ensemble-member mean CanESM2 precipitation averaged over 1977-2006, and (c) 100 ensemble-member mean MPI-ESM precipitation averaged over 1971-2000. PCC denotes the spatial correlation coefficient between simulations and observations.

Figure 2. Same as Figure 1, except that the shading denotes the annual range of precipitation (mm/day). The annual range is defined as the difference between local summer and winter precipitation. Local summer is May-September in the Northern Hemisphere and November-March in the Southern Hemisphere, and local winter is opposite. The climatological global monsoon domain is enclosed by black solid lines.

Figure 3. Changes in the annual range of precipitation (shading, mm/day) in CanEMS2, relative to the current climate, when the GMST is increased by 1.5 to 5 °C above the preindustrial level. Regions with gray slashes are the places where changes in the annual range of precipitation are statistically significant at the 0.05 confidence level, based on the student t-test. Purple dots denote that the results of more than 75% of the members are consistent. The climatological global monsoon domain is enclosed by black solid lines.

Figure 4. Same as Figure 3, except for changes in MPI-ESM.

Figure 5. Scatter plots of the relationships between (a-b) global monsoon area (10^6 km^2) and GMST, and (c-d) global monsoon intensity (mm/day) and GMST. Plots (a) and (c) are for CanESM2, and plots (b) and (d) are for MPI-ESM. Black solid lines denote model ensemble means with LOESS smoothing.

Figure 6. Smoothed histograms of projected global monsoon changes. (a) and (b) global monsoon area (10^6 km^2) , and (c) and (d) global monsoon intensity (mm/day). Plots (a) and (c) are for CanESM2, and plots (b) and (d) are for MPI-ESM. Black, purple, gray, green, orange, and red solid lines represent the histograms of the current climate and GMST increases of 1.5, 2, 3, 4, and 5°C above the preindustrial level, respectively.

Figure 7. (a) Percentage changes of global monsoon area (orange) and global monsoon intensity (green) per 1°C global warming (% /°C), derived from CanESM2 (left) and MPI-ESM (right). The upper (lower) edge of the box denotes the 75th (25th) percentile, and the horizontal line within the box is the ensemble mean. The vertical solid line denotes the range from 5th to 95th. (b)

- (d) Same as (a), except for global land monsoon, NH land monsoon, and SH land monsoon, respectively.

Figure 8. Same as Figure 7, except for regional land monsoons over (a) East Asia, (b) South Asia, (c) North Africa, (d) South Africa, (e) North America, (f) South America, and (g) Australia, respectively.

Figure 9. Scatter plots of relationships between warming levels (°C) and SAT differences (°C) between the equatorial eastern Pacific (5°S-5°N, 120°-80°W) and the tropical Atlantic (10°-20°N, 60°-15W) in CanESM2 (a) and MPI-ESM (b). (c-d) Same as (a-b), but for relationships between the North American land monsoon precipitation (mm/day) and SAT differences (°C) between the equatorial eastern Pacific and the tropical Atlantic. Red solid lines denote the linear regression. Numbers at the top-right denote the regression coefficients.

Figure 10. Moisture budget for precipitation changes $(10^9 \text{ m}^3/\text{day})$ of global (a) and regional (b-h) land monsoons with different global warming levels. Black, light-blue, blue, red, and pink bars denote the model-simulated precipitation changes (P), diagnosed precipitation changes (sum), dynamic component (DY), thermodynamic component (TH), and evaporation (E), respectively. Green solid lines denote the range of one standard deviation.